Author’s response to reviews

Title: Cytomegalovirus colitis in a patient with recurrent colon cancer after systemic chemotherapy: a case report

Authors:

Fuminori Teraishi (ftera@chikubageka.jp)
Hiroshi Shimamura (ftera@chikubageka.jp)
Takeo Suzuki (ftera@chikubageka.jp)
Masako Nakamoto (ftera@chikubageka.jp)
Akira Chikuba (ftera@chikubageka.jp)
Masashi Nezu (ftera@chikubageka.jp)
Shun Kohsaka (ftera@chikubageka.jp)
Takao Takiue (ftera@chikubageka.jp)
Hiroshi Chikuba (ftera@chikubageka.jp)

Version: 6 Date: 3 July 2008

Author’s response to reviews: see over
June 30, 2008

Michael Kidd, Editors-in-Chief,
Journal of Medical Case Reports

Dear Professor Kidd:

Thank you for your e-mail of June 26, 2008, regarding our manuscript (MS-7273690691656345). We also want to thank the deputy editor for providing helpful comments on the manuscript. The manuscript has been revised in response to all comments from the editor.

**Referee 1 (Deputy editor):**

Point 1: about CMV antigenemia  
Author’s response: We have confirmed the method for detection of CMV. It was detected by indirect enzyme antibody method (also known as antigenemia method).

Point 2: about adjuvant chemotherapy  
Author’s response: The patient was not received any adjuvant chemotherapies before FOLFIRI although we recommended him to do it after his operation. He presented the abdominal symptoms 2 weeks after initial chemotherapy consisting of FOLFIRI.

Point 3:  
Author’s response: We have changed the unit for chemotherapy dose from body into mg/m^2.

I have underlined the text parts that were changed in the revision. I hope that the revised manuscript now addresses all of the reviewers’ comments and that it can now be published in your prestigious journal. I am looking forward to hearing from you.

Thank you.

Sincerely,

Fuminori Teraishi